Author’s response to reviews

Title: Agreement Study between Color and IR Retinal Images Based on Retinal Vasculature Morphological Parameters

Authors:

Aqsa Ajaz (aqsa.ajaz@gmail.com)
Behzad Aliahmad (behzad.aliahmad@rmit.edu.au)
Himeesh Kumar (himeesh@gmail.com)
Marc Sarossy (marc@sarossy.com)
Dinesh Kumar (dinesh@rmit.edu.au)

Version: 3 Date: 02 Dec 2018

Author’s response to reviews:

The following changes have been made:

1. Materials section has been merged with Methods.

2. Statement that all authors have studied the manuscript and approve it.

3. The labels from all figures have been removed.

4. All Figure titles have been removed from the text and added to the list of Figures section, located after the Reference list.

To identify the location of the Figures in the text, only the word, 'Figure' has been typed at the location of the figure.

Thank you, Dinesh